Author's response to reviews

Title: Preoperative serum HER2 extracellular domain levels in primary invasive breast cancer

Authors:

Sae Byul Lee (newstar153@hanmail.net)
Jong Won Lee (jongwonlee116@gmail.com)
Jong Han Yu (lymbics@hanmail.net)
Beom Seok Ko (spdoctor@naver.com)
Hee Jeong Kim (haapybirth@amc.seoul.kr)
Byung Ho Son (brdrson@korea.com)
Gyung Yub Gong (gygong@amc.seoul.kr)
Hee Jin Lee (backilila@gmail.com)
Sung-Bae Kim (sbkim3@amc.seoul.kr)
Kyung Hae Jung (khjung@amc.seoul.kr)
Jin-Hee Ahn (ahnjiny@amc.seoul.kr)
Woochang Lee (wlee1@amc.seoul.kr)
Jooohon Sung (jsung@snu.ac.kr)
Sei-Hyun Ahn (ahnsh@amc.seoul.kr)

Version: 2
Date: 18 May 2014

Author's response to reviews:

I include an Authors’ Contributions section at the end of the manuscript,
I modify the manuscript about informed consent as below.

: Written informed consent was taken prior to surgery in all patients and the study protocol including the use of the database was approved by the Institutional Review Board of Asan Medical Center.